Author's response to reviews

Title: In vitro tooth whitening effect of two medicated chewing gums compared to a whitening gum and saliva

Authors:

  Michael Moore (moorem@ipfw.edu)
  Nathalie Hasler-Nguyen (nathalie.hasler-nguyen@novartis.com)
  Geoffrey Saroea (geoffrey.saroea@novartis.com)

Version: 8 Date: 6 August 2008

Author's response to reviews: see over
Title: In vitro teeth whitening effect of two nicotine medicated chewing gums compared to a whitening gum and saliva (MS: 2709242571870834)

Authors:
Michael Moore: moorem@ipfw.edu
Nathalie Hasler-Nguyen: nathalie.hasler-nguyen@novartis.com
Geoffrey Sareoa: geoffrey.saroea@novartis.com

Version: 6 Date: August 5th, 2008

Author’s response to reviews

To: Ashleigh Manning
Assistant Editor
BMC-series journals

Dear Mrs Manning,

We formatted table in the manuscript to comply to the journal format and we have made the change according to the editorial request.

We greatly appreciate your cooperation and look forward to hearing from you.

With best regards,

Nathalie Hasler-Nguyen, PhD

Formatting changes requested

Major revisions (we require the author to make these changes)

Table in parts - Table 1 has been divided into parts, a and b. It is important that tables are numbered in ascending numerical order - 1, 2 and 3. Can we please ask that the tables are renumbered and any references to the tables in the text are updated to reflect these changes? Alternatively, please merge the tables into one larger complete table.

Answer: table 1 and table 2 were created and reference to the tables in the text was updated.

Minor revisions (we can make these changes on the authors behalf)

Table title and legend The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.

Answer: This was done for table 1 and table 2
Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:
http://www.biomedcentral.com/info/ifora/figures

Answer: Cropping for Figure 4 was made.